\documentclass[a4paper,10pt]{article}

\usepackage{marvosym}
%\usepackage{fontspec} % xetex and fonts
\usepackage{url,parskip} % formatting
%\usepackage{xunicode,xltxtra} % other packages for xetex
 
% Graphics - Colors
\RequirePackage{color,graphicx}
\usepackage[usenames,dvipsnames]{xcolor}

% better formatting of the A4 page
\usepackage[empty]{fullpage}
 
\usepackage{supertabular} % for Grades
\usepackage{titlesec} % custom section
 
% Setup hyperref package, and colours for links
\usepackage{hyperref}
\definecolor{linkcolour}{rgb}{0,0.2,0.6}
\hypersetup{colorlinks,breaklinks,urlcolor=linkcolour,linkcolor=linkcolour}

% xetex & fonts
%\defaultfontfeatures{Mapping=tex-text}
%\setmainfont[SmallCapsFont = Fontin SmallCaps]{Fontin}

% CV Sections inspired by:
% http://stefano.italians.nl/archives/26
\titleformat{\section}{\large\scshape\raggedright}{}{0em}{}[\titlerule]
\titlespacing{\section}{0pt}{3pt}{3pt}

%-------BEGIN DOCUMENT--------
\begin{document}

\pagestyle{empty}

 %--------------------TITLE-------------
\par{\centering
{\Huge\textsc{Frank Zirkelbach} }
\bigskip\par}

\section{Personal Data}
\begin{tabular}{rl}
\textsc{Place and Date of Birth:} & Berlin, Germany  | 17 November 1977\\
\textsc{Address:} & R\"omerweg 10, 86391 Stadtbergen, Germany\\
\textsc{Phone:}	& +49 821 598 3008 (work)\\
	& +49 175 5228066 (mobile)\\
\textsc{Fax:}	& +49 821 598 3425 (work)\\
\textsc{email:}	& \href{mailto:frank.zirkelbach@physik.uni-augsburg.de}{frank.zirkelbach@physik.uni-augsburg.de}\\
\textsc{http:}	& \href{http://www.physik.uni-augsburg.de/~zirkelfr}{http://www.physik.uni-augsburg.de/\textasciitilde{}zirkelfr}\\
\end{tabular}
\begin{picture}(0,0)(-15,38)
%\includegraphics[height=3.0cm]{img/frank_app_ult.eps}
\includegraphics[height=3.0cm]{img/frank_app_mountain.eps}
\end{picture}

\section{Work Experience}
\begin{tabular}{r|p{11cm}}
\textsc{Jan 2006 - Present} & \textbf{Research assistant \& Doctoral candidate}\newline
                 Physics Department, University of Augsburg \\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Jul 2009 - Dec 2009} & \textbf{Research scientist}\newline
                 Physics Department, University of Paderborn\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\end{tabular}

\section{Basic education}
\begin{tabular}{r|p{11cm}}
\textsc{Sep 1988 - Jul 1998} & \textbf{Grammar school}\newline
                               Justus-von-Liebig-Gymnasium, Neus\"a\ss\newline
                               Holbein-Gymnasium, Augsburg\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Sep 1984 - Jul 1988} & \textbf{Elementary school}\newline
                               Ernst-Habermann-Grundschule, Berlin, Wilmersdorf\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\end{tabular}

\section{Conscription}
\begin{tabular}{r|p{11cm}}
\textsc{Aug 1998 - Sep 1999} & \textbf{Alternative civilian service}\newline
                               Hessing-Klinik, Augsburg\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\end{tabular}

\section{Scientific education}
\begin{tabular}{r|p{11cm}}
\textsc{Jan 2006 - Present} & \textbf{Doctoral studies in physics}\newline
                           Physics Department, University of Augsburg\newline
 {\small
 \begin{tabular}{lp{10cm}}
 Thesis: &
 {\em
 Atomistic simulation study on
 silicon carbide precipitation in silicon}\\
 \end{tabular}
 }\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Jul 2009 - Present} & \textbf{Scientific collaboration with the University of Paderborn}\newline
                 Theoretical Physics, Physics Department\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Jan 2006 - Dec 2008} & \textbf{Scholarship student}\newline
                           Bayerische Forschungsstiftung\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Oct/Nov 2007} & \textbf{Research period at the University of Helsinki}\\
\textsc{Aug/Sep 2008} & Division of Materials Physics, Department of Physics\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Oct 1999 - Dec 2005} & \textbf{Studies in physics}\newline
                               Physics Department, University of Augsburg\newline
 {\small
 \begin{tabular}{lp{10cm}}
 Thesis: &
 {\em
 Monte-Carlo-Simulation von selbstorganisierten nanometrischen\newline
 SiC$_x$-Ausscheidungen in C$^+$-implantierten Silizium}\\
 \end{tabular}
 }\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\end{tabular}

\addcontentsline{toc}{section}{List of publications}
\renewcommand{\refname}{List of publications}
\bibliography{../bibdb/bibdb}{}
\bibliographystyle{cv}
\nocite{zirkelbach11,zirkelbach10,zirkelbach09}
\nocite{zirkelbach2007,zirkelbach2006,zirkelbach2005}
\vspace*{0.3cm}

\section{Computer literacy}
\begin{tabular}{r|p{11cm}}
\textsc{Operating systems} & Linux, Unix, Windows\\
%\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Programming} & C (excellent), POSIX Threads, MPI, Fortran (reading \& modifying),\newline
                       RISC Assembler (Atmel, Mips), VHDL\\
%\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Scripting} & Bash, Sed, Awk, Make\\
%\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Revision control} & CVS, Git, SVN\\
%\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Text \& Layout} & LaTeX, GNUPlot, Gimp, XFig,
                          Inkscape, Openoffice\\
%\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\textsc{Others} & VASP (First-principles plane-wave based simulation package),\newline
                  Eagle (PCB Design), Xilinx ISE (VHDL Design), GDB\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\end{tabular}

\section{Knowledge of languages}
\begin{tabular}{r|p{11cm}}
\textsc{German} & Native language\\
\textsc{English} & Excellent (written \& spoken)\\
\textsc{French} & Basic knowledge\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\end{tabular}

\section{Leisure activities}
\begin{tabular}{r|p{11cm}}
\textsc{Sports} & Hiking, Trekking, Basketball, Volleyball\\
\textsc{Technical} & Programming, Blackbox reversing, Hardware and logic designs\\
\textsc{Reading} & Social-scientific texts\\
\multicolumn{2}{c}{}\\ %this clears the space between two jobs
\end{tabular}

\end{document}

